Author’s response to reviews

Title: The Relationship between Personality Traits and Marital Satisfaction: A Systematic Review and Meta-Analysis

Authors:

Kourosh Sayehmiri (sayehmiri@razi.tums.ac.ir)
Karez Ibrahim Kareem (Karez.shera@bnu.edu.iq)
Kamel Abdi (Kamel.abdi@komar.edu.iq)
Sahar Dalvand (S_dalvand90@yahoo.com)
Reza Ghanei Gheshlagh (rezaghanei30@yahoo.com)

Version: 3 Date: 27 Jan 2020

Author’s response to reviews:

Thank you for your comments. Our changes and explanations are as follows:

1. While assessing your manuscript it was noticed that Table 2 was not found to be mentioned in the main text. Please ensure that all tables and figures are referenced so as to provide context for the reader and also as an indication for where our production team is to insert them.
Answer: Thanks. We refer to Tables 1 and 2 in the text.

2. We note that only summarized data is presented in this manuscript, so we request that you amend this statement. Submission of a manuscript to a BMC Series journal implies that materials described in the manuscript, including all relevant raw data, will be freely available to any scientist wishing to use them for non-commercial purposes, without breaching participant confidentiality.
Answer: We have modified this section as follows. The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.

3. Currently, the contributions of authors Karez Ibrahim Kareem and Kamel Abdi do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.
Answer: As you may know, English is not good for most of Iranian researchers. So the two new authors were asked to collaborate to review the text. That's what the two did. If you do not agree, their names will be mentioned in the acknowledgement section.